% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{{\bf A combinatorial identity of multiple\\  zeta
values with even arguments}\thanks{This work was supported by
Natural Science Foundation of China (Nos.11101245, 61271355,
11271208,
 61202362, 11301302), China Postdoctoral Science
Foundation (No.2013M530869), Natural Science Foundation of
Shandong (No. BS2013SF009).  }}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address

\author{Shifeng Ding, Lihua Feng, Weijun Liu\\
\small Department of Mathematics\\[-0.8ex]
\small Central South University\\[-0.8ex]
\small Changsha 410083, China\\
\small\tt dingssff@163.com}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{December 1, 2013}{?? ??, 2014}{,}\\
\small Mathematics Subject Classifications: 05A19, 11M06}

\begin{document}
\begin{thebibliography}{99}

\bibitem{Gangl} H. Gangl, M. Kaneko, D. Zagier. Double zeta values and modular
forms. In {\em  Siegfried {\textnormal{(}Ed.\textnormal{)}},
Automorphic Forms and Zeta Functions}.  In Memory of Tsuneo
Arakawa, Proc. of the Conf., Rikkyo University, Tokyo, Japan, 4-7
September 2004, pages 71-106. World Scientific B\"{o}herer,
Hackensack, NJ, 2006.



\bibitem{Granville} A. Granville. \newblock A
decomposition of Riemann's zeta function.\newblock  In\ {\em
Analytic Number Theory}, Volume 247 of {\em London Math. Soc.
Lecture Note Ser.,}  pages 95-101. Cambridge University Press,
1997.

\bibitem{Hoffman4} M. E. Hoffman. \newblock Multiple harmonic series.\newblock
{\em Pacific J. Math}, {152(2): 275-290}, 1992.

\bibitem{HoffmanAndMoen} M. E. Hoffman, C. Moen. \newblock Sums of triple
harmonic series. \newblock {\em J. Number Theory}, {60(2):
329-331}, 1996 .

\bibitem{Hoffman} M. E. Hoffman. \newblock On multiple zeta values of even arguments.\newblock
{\em arXiv}:\ 1205.7051.



\bibitem{Masahiro Igarashi} M. Igarashi. \newblock On generalizations of the sum formula
for multiple zeta values.\newblock{\em arXiv}: 1110.4875.


\bibitem{Kaneko} M. Kaneko. Introduction to multiple zeta values,
(in Japanese). In {\em Algebraic number theory and related topics.
}  ( Kyoto, 1998), pages 50-68. S\={u}rikaisekikenky\={u}sho
K\={o}ky\={u}roku 1097, 1999.
\bibitem{Y. Liu}Y. Liu.\newblock  Cylinder functions (in Chinese)\newblock, National Defence
Industrial Press, Bejing, 1983.


\bibitem{Markett} C. Markett. \newblock Triple sums and the Riemann zeta function.\newblock
{\em J. Number
Theory}, {48(2): 113-132}, 1994.

\bibitem{Remmert}R. Remmert. Classical topics in complex function
theory, GTM172, Springer, 2013.

\bibitem{Z.Y.Shen} Z. Y. Shen, T. X. Cai.\newblock  Some
identities for multiple zeta values. \newblock {\em J.Number
Theory}, {132: 314-323}, 2012.


\bibitem{Zhao} J. Q. Zhao,\newblock  Sum Formula of Multiple Hurwitz-Zeta
Values.\newblock {\em  arXiv}: 1207.2368.




\end{thebibliography}


\maketitle

% E-JC papers must include an abstract. The abstract should consist of a
% succinct statement of background followed by a listing of the
% principal new results that are to be found in the paper. The abstract
% should be informative, clear, and as complete as possible. Phrases
% like "we investigate..." or "we study..." should be kept to a minimum
% in favor of "we prove that..."  or "we show that...".  Do not
% include equation numbers, unexpanded citations (such as "[23]"), or
% any other references to things in the paper that are not defined in
% the abstract. The abstract will be distributed without the rest of the
% paper so it must be entirely self-contained.

\begin{abstract}
Let $\zeta(s_1,s_2,\cdots,s_k;\alpha)$ be the multiple Hurwitz
zeta function. Given two positive integers $k$ and $n$ with $k\leq
n$, let $E(2n, k;\alpha)$ be the sum of all multiple zeta values
with even arguments whose weight is $2n$ and whose depth is $k$.
In this note we  present some generating series for the numbers
$E(2n,k;\alpha)$.

% keywords are optional
\bigskip\noindent \textbf{Keywords:}\ Multiple zeta values;  Recursion algorithm; Generating
series
\end{abstract}

\section{Introduction}
The multiple zeta values
\begin{equation}\zeta(s_1,s_2,\cdots,s_m)=\sum_{n_1>n_2>\cdots>n_k>0}
\frac{1}{n_1^{s_1}n_2^{s_2}\cdots n_m^{s_m}},\nonumber
\end{equation}
are also called Euler-Zagier sums, where $s_1$, $s_2$, $\cdots$, $
s_m$ are positive integers with $s_1\geq2$. The multiple Hurwitz
zeta function $\zeta(s_1,s_2,\cdots,s_k;\alpha)$ is defined by the
multiple series
\[\zeta(s_1,s_2,\cdots,s_k;\alpha)=\sum_{n_1>n_2>\cdots>n_k\geq0}
\frac{1}{(n_1+\alpha)^{s_1}(n_2+\alpha)^{s_2}\cdots
(n_k+\alpha)^{s_k}},\] where
$\alpha\in\mathbb{C}\setminus\{0,-1,-2,\cdots\}$ and
$s_1,s_2,\cdots,s_k$ are positive integers with $s_1>1$. The
multiple zeta functions have attracted considerable interest in
recent years.

There are various relations among multiple zeta values. One of the
well-known Q-linear relations among multiple zeta values is the
sum formula\ ( see\
$\cite{Granville,Hoffman4,HoffmanAndMoen,Kaneko,Markett}$), which
states that
\[\sum_{\stackrel{j_1+\cdots+j_k=n}{\textnormal{\scriptsize {Each}}j_i\geq1,j_1>1}}
\zeta(j_1,j_2,\cdots,j_k)=\zeta(n).\]

In \cite{Masahiro Igarashi}, M. Igarashi proved a generalization
of the sum formula. Here we quote Igarashi's result in a slightly
different form.
\begin{theorem}{\textnormal{(\cite{Masahiro Igarashi})}}\ \ Let $\alpha$ be a complex number with
$\textnormal{Re}\ \alpha>0$ and let $n,k$ be positive integers
such that $n\geq k+1$. Then the following identity holds.
\begin{eqnarray}&&\sum_{\stackrel{j_1+j_2+\cdots+j_k=n}{\textnormal{Each}j_i\geq1,
j_1\geq2}}\zeta(j_1,j_2,\cdots,j_k;\alpha)\nonumber\\
&=&\frac{1}{(n-k-1)!}\sum_{N=0}^\infty\frac{1}{(N+\alpha)^k}\left.\left[\frac{(1-x)_{N}}
{(\alpha-x)_{N+1}}\right]^{(n-k-1)}\right|_{x=0},\nonumber
\end{eqnarray}
where $[\ \cdot \ ]^{(n-k-1)}$ denotes the $(n-k-1)$th derivative
of a function with respect to its variable $x$, and $(a)_N$
denotes the Pochhammer symbol.
\end{theorem}

Given two positive integers $n$ and $k$ with $n\geq k$, let $E(2n,
k;\alpha)$ be the sum of all the multiple zeta values of
even-integer arguments having weight $2n$ and depth $k$, i.e.,
\[E(2n,k;\alpha)=\sum_{\stackrel{j_1+\cdots+j_k=n}{j_1,j_2,\cdots,j_k\geq1}}
\zeta(2j_1,2j_2,\cdots,2j_k;\alpha).\] (The notation
$E(2n,k;\alpha)$ is similar to that in  \cite{Hoffman, Zhao}. In
\cite{Hoffman}, Hoffman defined a number$E(2n,k)$ which is
equivalent to the number $E(2n,k;1)$ in our notation, and in
\cite{Zhao}, the number $E(2n,k;1/2)/2^{2n}$ in our notation is
denoted by $T(2n,k)$.)

The result of Gangl, Kaneko and Zagier \cite{Gangl} that
\begin{eqnarray}E(2n,2;1)=\frac{3}{4}\zeta(2n), \ \ {\textnormal{for }}\
n\geq2.\nonumber
\end{eqnarray}
was extended by Shen and Cai \cite{Z.Y.Shen} to

\begin{equation}E(2n,3;1)=\frac{5}{8}\zeta(2n)-\frac{1}{4}\zeta(2)\zeta(2n-2),\
\ \  {\textnormal{for }}n\geq3,\nonumber
\end{equation}
\begin{equation}E(2n,4;1)=\frac{35}{64}\zeta(2n)-\frac{5}{16}\zeta(2)\zeta(2n-2),\
\ \ {\textnormal{for }}n\geq4.\nonumber
\end{equation}

Applying the theory of symmetric functions, Hoffman \cite{Hoffman}
and Zhao \cite{Zhao}, respectively, established the generating
functions for the numbers $E(2n,k;1)$ and $T(2n,k)$, respectively.
Hoffman \cite{Hoffman} proved that
\[1+\sum_{n\geq k\geq1}E(2n,k;1)t^ns^k=\frac{\sin(\pi\sqrt{1-s}\sqrt{t})}{\sqrt{1-s}\sin(\pi\sqrt{t})}.\]
And Zhao \cite{Zhao} proved that
\[1+\sum_{n\geq k\geq1}T(2n,k)t^ns^k=\frac{\cos(\pi\sqrt{(1-s)t}/2)}{\cos(\pi\sqrt{t}/2)}.\]
Based on these generating functions, some  formulas for the
numbers $E(2n,k;1)$ and $T(2n,k)$ for arbitrary $n\geq k$ have
been obtained. For example,
\begin{eqnarray}E(2n,k;1)=\frac{\zeta(2n)}{2^{2(k-1)}}{2k-1\choose k}-
\sum_{j=1}^{[\frac{k-1}{2}]}\frac{\zeta(2j)\zeta(2n-2j)}{2^{2k-3}(2j+1)B_{2j}}{2k-2j-1\choose
k}, \ \ \ \nonumber\end{eqnarray}
\[T(2n,k)=\frac{t(2n)}{2^{2(k-1)}k}{2k-2\choose k-1}-
\sum_{j=1}^{\lfloor\frac{k-1}{2}\rfloor}\frac{t(2j)t(2n-2j)}{2^{2k-3}(2^{2j}-1)kB_{2j}}{2k-2j-2\choose
k-1},\]
 where $B_{2j}$ is the $2j$th Bernoulli number.


In this note we use a method introduced by Granville
\cite{Granville} to establish an algorithm similar to that in
Theorem 1 for the calculation of the numbers $E(2n,k;\alpha)$. We
prove that the calculation of $E(2n,k;\alpha)$ involves the Euler
$\Gamma$-function and the direct formulas for $E(2n,k;1)$ and
$E(2n,k;1/2)$ can be deduced from the Bessel functions of the
first kind. The main results of this paper are the following
theorems.


\begin{theorem}\ Let $\alpha$ be a complex number with
$\alpha\in\mathbb{C}\setminus\{0,-1,-2,\cdots\}$. Let $N\geq0$
denote an integer and let $(a_0^{(N)},a_1^{(N)},a_2^{(N)},\cdots)$
 be an infinite series defined by the expansion
\begin{equation}\prod_{\stackrel{r=0}{r\not=N}}^\infty\left(1+\frac{x}{(r+\alpha)^2-(N+\alpha)^2}\right)=a_0^{(N)}+a_1^{(N)}x+\cdots+a_{k-1}^{(N)}x^{k-1}+\cdots.
\end{equation}
Then for positive integers $n$ and $k$ (suppose $n\geq k$), we
have
\begin{equation}E(2n,k;\alpha)=\sum_{N=0}^\infty\frac{a_{k-1}^{(N)}}{(N+\alpha)^{2n-2k+2}}.\end{equation}
\end{theorem}



\begin{theorem}\ Given an integer $N\geq0$ and a complex number
$\alpha\in\mathbb{C}\setminus\{0,-1,-2,\cdots\}$, for any
$x\in\mathbb{C}$  we have
\begin{eqnarray}&&\prod_{\stackrel{r=0}{r\not=N}}^\infty\left(1+\frac{x}{(r+\alpha)^2-(N+\alpha)^2}\right)\nonumber\\
&=&\frac{2(-1)^{N}(N+\alpha)\Gamma(2\alpha+N)}{N!}\frac{1}{x\Gamma(\alpha+\sqrt{(N+\alpha)^2-x})\Gamma(\alpha-\sqrt{(N+\alpha)^2-x})},
\nonumber
\end{eqnarray}
where $\Gamma$ denotes Euler's Gamma function.

\end{theorem}


\begin{corollary}\ Theorem 2 and Theorem 3 yield that: \begin{eqnarray}&&E(2n,k;\alpha)\nonumber\\
&=&\frac{2}{k!}\sum_{N=0}^\infty\frac{(-1)^N\Gamma(N+2\alpha)}{N!(N+\alpha)^{2n-2k+1}}
\left.\left[\frac{1}{\Gamma(\alpha+\sqrt{(N+\alpha)^2-x})
\Gamma(\alpha-\sqrt{(N+\alpha)^2-x})}\right]^{(k)}\right|_{x=0},\nonumber
\end{eqnarray}
where $[\ \cdot \ ]^{(k)}$ denotes the $k$th derivative of a
function with respect to its variable $x$ and\ $\Gamma$  denotes
the Gamma function.
\end{corollary}

\begin{corollary}\ For positive integers $n $ and $k$ (suppose $n\geq k$ ), we have
\begin{eqnarray}E(2n,k;1)&=&\frac{4(-1)^{k+1}\pi^{2k}}{k!}\sum_{N=0}^\infty
\frac{(-1)^N}{(N+1)^{2n-2k}}\left(\cos\sqrt{x}\right)^{(k+1)}|_{x=(N\pi+\pi)^2},\\
E(2n,k;\frac{1}{2})&=&\frac{2(-1)^{k}\pi^{2k-1}}{k!}\sum_{N=0}^\infty
\frac{(-1)^N}{(N+\frac{1}{2})^{2n-2k+1}}\left(\cos\sqrt{x}\right)^{(k)}|_{x=(N\pi+\pi/2)^2}.
\end{eqnarray}
\end{corollary}

\begin{corollary}\ The direct formulas for $E(2n,k;1)$ and
$E(2n,k;\frac{1}{2})$ are that:
\begin{eqnarray}&&E(2n,k;1)=\frac{1}{2^{2k-2}k!}
\sum_{j=0}^{\lfloor\frac{k-1}{2}\rfloor}\frac{(-1)^{j}(2k-1-2j)!(2\pi)^{2j}}{(2j+1)!(k-1-2j)!}\cdot\zeta(2n-2j),\\
&&E(2n,k;\frac{1}{2})=\frac{1}{2^{2k-2}k!}\sum_{j=0}^{\lfloor\frac{k-1}{2}\rfloor}\frac{(-1)^j(2k-2-2j)!(2\pi)^{2j}}{(2j)!(k-1-2j)!}
\zeta(2n-2j;\frac{1}{2}).
\end{eqnarray}
\end{corollary}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Some Lemmas}

This paper uses some properties of Euler's Gamma function
$\Gamma(s)$ and  the Bessel function $J_{p}(x)$ of the first kind.
Here we list some needed Lemmas.

There are several equivalent forms of the Gamma function
$\Gamma(s)$ developed by Euler, one by Weierstrass:

\[\frac{1}{\Gamma(s)}=se^{\gamma s}\prod_{m=1}^\infty\left(1+\frac{s}{m}\right)e^{-\frac{s}{m}},\ \ \forall s\in\mathbb{C},\]
where $\gamma$ denotes the Euler-Mascheroni constant defined by
$$\gamma=\lim_{m\rightarrow\infty}(1+\frac{1}{2}+\cdots+\frac{1}{m}-\log
m)=0.577215\cdots.$$ It is easy to deduce from Weierstrass'
definition of $\Gamma(s)$ the following lemma.

\begin{lemma}\ Let $\alpha_1,\alpha_2,\cdots,\alpha_k$ and
$\beta_1,\beta_2,\cdots,\beta_k$ be two groups of complex numbers
\textnormal{($k\geq1$)}. Suppose they satisfy the following
conditions{\textnormal{:\ (i)}}\
$\alpha_1+\alpha_2+\cdots+\alpha_k=\beta_1+\beta_2+\cdots+\beta_k$
and{\textnormal{ (ii)}} none of the $\beta_j$s is a negative
integer. Then we have
\[\prod_{m=1}^\infty\frac{(m+\alpha_1)(m+\alpha_2)\cdots(m+\alpha_k)}{(m+\beta_1)(m+\beta_2)\cdots(m+\beta_k)}
=\frac{\Gamma(1+\beta_1)\Gamma(1+\beta_2)\cdots\Gamma(1+\beta_k)}{\Gamma(1+\alpha_1)\Gamma(1+\alpha_2)\cdots\Gamma(1+\alpha_k)}.
\]
\end{lemma}

It is well known that $\Gamma(s)$ satisfies the functional
relation $\Gamma(s+1)=s\Gamma(s), \ \forall
s\in\mathbb{C}\setminus\{0,-1,-2,\cdots\}$, and if $N\geq0$ is an
integer then $\Gamma(s)$ has residue $\frac{(-1)^N}{N!}$ at
$s=-N$.

\begin{lemma}\textnormal{\cite{Remmert}} The following identities
hold for $s\in\mathbb{C}$:
\[\frac{1}{\Gamma(1+s)\Gamma(1-s)}=\frac{\sin\pi s}{\pi s},\ \ \ \
\ \ \
\frac{1}{\Gamma(\frac{1}{2}+s)\Gamma(\frac{1}{2}-s)}=\frac{\cos\pi
s}{\pi } .\ \]
\end{lemma}



We turn to introduce some properties of the  Bessel function of
the first kind with a half-integer index. The Bessel function
$J_p(x)$ is said to be of a half integer index if $p=
k\pm\frac{1}{2}$ with $k$ being an integer. It is well known that
the Bessel function of the first kind with a half integer index
can be represented by elementary functions.

\begin{lemma}\ \ $\frac{d^k}{dx^k}\cos\left(\sqrt{x}\right)=\sqrt{\frac{\pi}{2}}(-1)^k2^{-k}x^{\frac{1-2k}{4}}
J_{k-\frac{1}{2}}(\sqrt{x}), \ \ \textnormal{for} \ x>0\
\textnormal{and}\ \ k=0,1,2,\cdots.$
\end{lemma}

\begin{proof}\ The Bessel function $J_p(x)$ is defined by the
series
\begin{equation}J_p(x)=\left(\frac{x}{2}\right)^{p}\sum_{m=0}^\infty\frac{(-1)^m}{m!\Gamma(p+m+1)}\left(\frac{x}{2}\right)^{2m},\end{equation}
where the radius of convergence of the series is $+\infty$. It
follows from (7) that  (using the well-known identity
$\Gamma(n+1/2)=\sqrt{\pi}(2n)!/(n!2^{2n})$, $\forall n\geq0$)

\begin{equation}J_{k-\frac{1}{2}}(\sqrt{x})=\sqrt{\frac{2}{\pi}}2^kx^{\frac{2k-1}{4}}\sum_{m=0}^\infty\frac{(-1)^m(m+k)!}
{m!(2m+2k)!}x^m.\end{equation}

On the other hand, by Taylor's expansion of $\cos\sqrt{x}$, we
have
\begin{eqnarray}\frac{d^k}{dx^k}\cos\left(\sqrt{x}\right)=\sum_{m=k}^\infty\frac{(-1)^mm(m-1)\cdots(m-k+1)}
{(2m)!}x^{m-k}=\sum_{m=0}^\infty\frac{(-1)^{m+k}(m+k)!}
{(2m+2k)!m!}x^{m}.\end{eqnarray} Observing the right-hand sides of
(8) and (9), the result follows immediately.
\end{proof}


\begin{lemma}\ Let
$k\geq0$ be an integer and let  $x>0$. Then the Bessel function
$J_{k+1/2}(x)$ is represented by

\begin{eqnarray}J_{k+\frac{1}{2}}(x)&=&\sqrt{\frac{2}{\pi }}\frac{1}{\sqrt{x}}\left\{\sin x
\sum_{j=0}^{\lfloor\frac{k}{2}\rfloor}\frac{(-1)^j(2k-2j)!}{(2j)!(k-2j)!}\cdot\frac{1}{(2x)^{k-2j}}\right.\nonumber\\
&&\left.\ \ \ \ \ \ \ \ \ \ \ \ -\cos x
\sum_{j=0}^{\lfloor\frac{k-1}{2}\rfloor}\frac{(-1)^j(2k-1-2j)!}{(2j+1)!(k-1-2j)!}\cdot\frac{1}{(2x)^{k-1-2j}}\right\}.
\end{eqnarray}
\end{lemma}

\begin{proof}\ \ We omit the detailed steps. It is shown that\ (\cite{Y. Liu}) the
Bessel function $J_{k+1/2}(x)$  satisfies the relation
\begin{equation}J_{k+1/2}(x)=(-1)^k\sqrt{\frac{2}{\pi
x}}x^{k+1}\frac{d^k}{(xdx)^k}\left(\frac{\sin x}{x}\right),\ \
\textnormal{for}\ \ k=0,1,2,\cdots.\end{equation} It follows from
(11) that
\[J_{1/2}=\sqrt{\frac{2}{\pi
}}\frac{\sin x}{\sqrt{x}},\ \ \ \ \ J_{3/2}(x)=\sqrt{\frac{2}{\pi
}}\frac{1}{\sqrt{x}}\left(\frac{\sin x}{x}-\cos x\right),\cdots,\]
and so on. Then using induction on $k$ we can prove that (10) is
equivalent to (11).\end{proof}



\begin{lemma}Let $k\geq1$ be an integer. Then by Lemma 9 and Lemma 10, we have
\begin{eqnarray}\left(\cos{\sqrt{x}}\right)^{(k)}&=&\frac{(-1)^{k}}{2^{2k-1}}\frac{\sin\sqrt{x}}{\sqrt{x}}
\sum_{j=0}^{\lfloor\frac{k-1}{2}\rfloor}\frac{(-1)^j(2k-2-2j)!2^{2j}}{(2j)!(k-1-2j)!}\cdot\frac{1}{x^{k-1-j}}\nonumber\\
&& -\frac{(-1)^{k}}{2^{2k-2}}\cos \sqrt{x}
\sum_{j=0}^{\lfloor\frac{k-2}{2}\rfloor}\frac{(-1)^j(2k-3-2j)!2^{2j}}{(2j+1)!(k-2-2j)!}\cdot\frac{1}{x^{k-1-j}}.
\end{eqnarray}
\end{lemma}





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

 \section{ Proofs}
\textnormal{(}{\it Proof of Theorem 2 }\textnormal{)}

\begin{proof}\ The left side of (2) is


\begin{eqnarray}
&&\sum_{\stackrel{j_1+\cdots+j_k=n}{j_1,\cdots,j_k\geq1}}\sum_{n_1>n_2>\cdots>n_k\geq0}
\frac{1}{(n_1+\alpha)^{2j_1}(n_2+\alpha)^{2j_2}\cdots(n_k+\alpha)^{2j_k}}\nonumber\\
&&\ \ \ \
=\sum_{n_1>n_2>\cdots>n_k\geq0}\sum_{\stackrel{j_1+\cdots+j_k=n}{j_1,\cdots,j_k\geq1}}
\frac{1}{(n_1+\alpha)^{2j_1}(n_2+\alpha)^{2j_2}\cdots(n_k+\alpha)^{2j_k}}.\end{eqnarray}




The second sum in (13) is the coefficient of $x^{2n}$ in the
formal power series
\begin{eqnarray}&&\sum_{j=1}^\infty\left(\frac{x}{n_1+\alpha}\right)^{2j}
\sum_{j=1}^\infty\left(\frac{x}{n_2+\alpha}\right)^{2j}\cdots\sum_{j=1}^\infty\left(\frac{x}{n_k+\alpha}\right)^{2j}\nonumber\\
&&\ \
=\frac{x^{2k}}{[(n_1+\alpha)^2-x^2][(n_2+\alpha)^2-x^2]\cdots[(n_k+\alpha)^2-x^2]}\nonumber\\
&&\ \
=\sum_{j=1}^{k}\left(\frac{x^{2k}}{(n_j+\alpha)^2-x^2}\prod_{\stackrel{1\leq
m\leq k}{m\not=j}}\frac{1}{(n_m+\alpha)^2-(n_j+\alpha)^2}\right).
\end{eqnarray}
It follows that the coefficient of $x^{2n}$ above is
\[\sum_{j=1}^k\left(\frac{1}{(n_j+\alpha)^{2n-2k+2}}\prod_{\stackrel{1\leq
m\leq
k}{m\not=j}}\frac{1}{(n_m+\alpha)^2-(n_j+\alpha)^2}\right).\]
Hence the sum (13) is
\begin{equation}\sum_{n_1>n_2>\cdots>n_k\geq0}\sum_{j=1}^k\left(\frac{1}{(n_j+\alpha)^{2n-2k+2}}\prod_{\stackrel{1\leq
m\leq k}{m\not=j}}\frac{1}{(n_m+\alpha)^2-(n_j+\alpha)^2}\right).
\end{equation}

Now  we take each $n_j=N$ in turn. Then the sum (15) becomes that

\begin{equation}\sum_{N\geq0}\frac{1}{(N+\alpha)^{2n-2k+2}}\sum_{j=1}^k
P(N,j-1)Q(N,k-j),
\end{equation}
where the series $\{P(N,i)\}_{i=0}^\infty$ and
$\{Q(N,i)\}_{i=0}^\infty$ are defined in the following manner.
Define $P(N,0)=Q(N,0)\triangleq1$, and $Q(N,j)=0$, $\forall j> N$;
If $j>1$ then define $P(N,j-1)$ to be
\begin{equation}P(N,j-1)=\sum_{n_1>\cdots>n_{j-1}>N}\prod_{i=1}^{j-1}\frac{1}{[(n_i+\alpha)^2-(N+\alpha)^2]};\nonumber\end{equation}
While if  $1\leq j<k $ and $k-j\leq N$ , then define $Q(N,k-j)$ to
be
\begin{equation}Q(N,k-j)=\sum_{N>n_{j+1}>\cdots>n_k\geq0}\prod_{i=j+1}^{k}\frac{1}{[(n_i+\alpha)^2-(N+\alpha)^2]}.\nonumber\end{equation}
In other words, the series $\{P(N,i)\}_{i=0}^\infty$ and
$\{Q(N,i)\}_{i=0}^\infty$ are defined by  the following generating
functions:
\[1+P(N,1)x+P(N,2)x^2+\cdots=\prod_{r>N}\left(1+\frac{x}{(r+\alpha)^2-(N+\alpha)^2}\right),\]
and \[1+Q(N,1)x+Q(N,2)x^2+\cdots=\prod_{0\leq
r<N}\left(1+\frac{x}{(r+\alpha)^2-(N+\alpha)^2}\right).\]

Now the second sum \ $\sum_{j=1}^kP(N,j-1)Q(N,k-j)$ in (16) is the
coefficients of $x^{k-1}$ in the power series
\[[1+P(N,1)x+P(N,2)x^2+\cdots]\cdot[1+Q(N,1)x+Q(N,2)x^2+\cdots],\]
i.e.,
\[\sum_{j=1}^kP(N,j-1)Q(N,k-j)=a_{k-1}^{(N)},\]
where $a_{k-1}^{(N)}$ is given by (1).

Therefore, the sum (16) is
\[\sum_{N\geq0}\frac{a_{k-1}^{(N)}}{(N+\alpha)^{2n-2k+2}},\]
which gives the identity (2).
\end{proof}


\noindent\textnormal{(}{\it Proof of Theorem 3 }\textnormal{)}

\begin{proof}\ \ Consider the infinite product
\[\prod_{r=0}^\infty\left(1+\frac{x}{(r+\alpha)^2-y^2}\right),\]
where $y$ is a complex number with
$y\not\in\{\alpha,1+\alpha,2+\alpha,\cdots\}$.

By Lemma 7, we have
\begin{eqnarray}&&\prod_{r=0}^\infty\left(1+\frac{x}{(r+\alpha)^2-y^2}\right).\nonumber\\
&=&\frac{\alpha^2-y^2+x}{\alpha^2-y^2}\prod_{r=1}^\infty\frac{(r+\alpha+\sqrt{y^2-x})
(r+\alpha-\sqrt{y^2-x})}{(r+\alpha+y)(r+\alpha-y)}\nonumber\\
&=&\frac{\alpha^2-y^2+x}{\alpha^2-y^2}\frac{\Gamma(1+\alpha+y)\Gamma(1+\alpha-y)}{\Gamma(1+\alpha+\sqrt{y^2-x})\Gamma(1+\alpha-\sqrt{y^2-x})}\nonumber\\
&=&\frac{\Gamma(\alpha+y)\Gamma(\alpha-y)}{\Gamma(\alpha+\sqrt{y^2-x})\Gamma(\alpha-\sqrt{y^2-x})}.
\end{eqnarray}


 It follows that
\begin{eqnarray}\prod_{\stackrel{r=0}{r\not=N}}^\infty\left(1+\frac{x}{(r+\alpha)^2-y^2}\right)=
\frac{\Gamma(\alpha+y)\Gamma(\alpha-y)}{\Gamma(\alpha+\sqrt{y^2-x})\Gamma(\alpha-\sqrt{y^2-x})}\times\frac{(N+\alpha)^2-y^2}{(N+\alpha)^2-y^2+x}.
\end{eqnarray}
By the residue formula of the Gamma function, we have
\[\lim_{y\rightarrow
N+\alpha}\Gamma(\alpha-y)(N+\alpha-y)=\lim_{s\rightarrow-N}\Gamma(s)(s+N)=\frac{(-1)^N}{N!}.\]
Now on both sides of (18) we put $y\rightarrow N+\alpha$. Then we
get the required  identity in Theorem 3.
\end{proof}

\noindent\textnormal{(}{\it Proof of Corollary 5 and Corollary 6
}\textnormal{)}

\begin{proof}\ \ In Corollary 4, we take $\alpha=1$ and $\alpha=1/2$,
respectively. By Lemma 8, we have
\[\frac{1}{\Gamma(1+\sqrt{(N+1)^2-x})
\Gamma(1-\sqrt{(N+1)^2-x})}=\frac{\sin(\pi\sqrt{(N+1)^2-x})}{\pi\sqrt{(N+1)^2-x}},\]

\[\frac{1}{\Gamma(\frac{1}{2}+\sqrt{(N+\frac{1}{2})^2-x}\ )
\Gamma(\frac{1}{2}-\sqrt{(N+\frac{1}{2})^2-x}\
)}=\frac{1}{\pi}\cos\left(\pi\sqrt{(N+\frac{1}{2})^2-x}\right).\]
It is easy to verify that
\begin{eqnarray}&&\left.\left(\frac{\sin\left(\pi\sqrt{(N+1)^2-x}\right)}{\pi\sqrt{(N+1)^2-x}}\right)^{(k)}\right|_{x=0}
=(-1)^{k}\pi^{2k}\left.\left(\frac{\sin\sqrt{x}}{\sqrt{x}}\right)^{(k)}\right|_{x={(N\pi+\pi)^2}}\nonumber\\
&=&2(-1)^{k+1}\pi^{2k}\left.\left(\cos\sqrt{x}\right)^{(k+1)}\right|_{x={(N\pi+\pi)^2}},
\end{eqnarray}

\begin{eqnarray}&&\left.\left(\cos(\pi\sqrt{(N+\frac{1}{2})^2-x}\ )\right)^{(k)}\right|_{x=0}
=(-1)^{k}\pi^{2k}\left.(\cos\sqrt{x})^{(k)}\right|_{x={(N\pi+\pi/2)^2}}.
\end{eqnarray}

Substitute the right-hand side of (19) into  the right-hand side
of the identity in Corollary 4, then we get the identity (3).
Similarly, we can prove (4).

To prove Corollary 6, we use Lemma 11. It is deduced from (12)
that
\begin{eqnarray}&&\left.\left(\cos\sqrt{x}\right)^{(k+1)}\right|_{x={(N\pi+\pi)^2}}\nonumber\\
&&=\frac{(-1)^{(N+k+1)}}{(2\pi)^{2k}}\sum_{j=0}^{\lfloor\frac{k-1}{2}\rfloor}\frac{(-1)^j(2k-1-2j)!}{(2j+1)!(k-2j-1)!}\frac{(2\pi)^{2j}}{(N+1)^{2k-2j}},
\end{eqnarray}
\begin{eqnarray}&&\left.\left(\cos\sqrt{x}\right)^{(k)}\right|_{x={(N\pi+\pi/2)^2}}\nonumber\\
&&=\frac{(-1)^{(N+k)}}{(2\pi)^{2k-1}}\sum_{j=0}^{\lfloor\frac{k-1}{2}\rfloor}
\frac{(-1)^j(2k-2-2j)!}{(2j)!(k-2j-1)!}\frac{(2\pi)^{2j}}{(N+\frac{1}{2})^{2k-1-2j}}.
\end{eqnarray}
Substitute (21) and (22) into (3) and (4) respectively, then we
will get the desired identities (5) and (6).
\end{proof}

\section{Generalizations}

Similar to the definition of $E(2n,k;\alpha)$, we can define a
number $E(mn,k;\alpha)$ as follows. Let $m,n$ and $k$ be positive
integers with $m\geq2$ and $n\geq k$. Then define the number
$E(mn,k;\alpha)$ to be
\[E(mn,k;\alpha)=\sum_{\stackrel{j_1+\cdots+j_k=n}{j_1,j_2,\cdots,j_k\geq1}}
\zeta(mj_1,mj_2,\cdots,mj_k;\alpha).\]

If we check the proofs of Theorem 2 and Theorem 3 again, we may
find that the theorems can be generalized in the following way.
\begin{theorem} Let $\alpha$ be a complex number with
$\alpha\not\in\{0,-1,-2,\cdots\}$ and let $N\geq0$ denote an
arbitrary integer. Define a function $f_{N}(x)$ by
\[f_N(x)=\prod_{\stackrel{r=0}{r\not=N}}^\infty\left(1+\frac{x}{(r+\alpha)^m-(N+\alpha)^m}\right).
\]
Then for positive integers $n$ and $k$ (suppose $n\geq k$), we
have
\[E(mn,k;\alpha)=\frac{1}{(k-1)!}\sum_{N=0}^\infty\frac{[f_N(x)]^{(k-1)}|_{x=0}}{(N+\alpha)^{mn-mk+m}}.\]
Moreover, the function $f_N(x)$ can be represented by the Gamma
function, i.e.,
\[f_N(x)=\frac{(-1)^Nm(N+\alpha)^{m-1}}{N!}\cdot\frac{1}{x}\cdot
\frac{\prod_{i=1}^{m-1}\Gamma(\alpha-\theta_i(N+\alpha))}{\prod_{i=1}^m\Gamma(\alpha-\theta_i{\sqrt[m]{(N+\alpha)^m-x}})},\]
where $\theta_1$, $\theta_2$, $\cdots$, $\theta_m(\triangleq1)$
are all the pairwise different roots of  $Z^m=1$ in $\mathbb{C}$.
\end{theorem}
\subsection*{Acknowledgements}
The authors would like to thank the referee for helpful
suggestions.




























%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% \bibliographystyle{plain}
% \bibliography{myBibFile}
% If you use BibTeX to create a bibliography
% then copy and past the contents of your .bbl file into your .tex file





\end{document}

\begin{thebibliography}{10}



\bibitem{bue}F. Buekenhout, A. Delandtsheer, J. Doyen, P. Kleidman,
M.Liebeck, and J. Saxl. \newblock Linear spaces with
flag-transitive automorphism groups. \newblock{\em Gemo.
Dedicata}., {36: 89-94}, 1990.

\bibitem{cam1}A. Camina and J. Siemons. \newblock Block transitive automorphism
groups of 2-$(v,k,1)$ block designs. \newblock {\em J.
Combinatorial Theory Ser. A}, {51: 268-276}, 1989.

\bibitem{cam2}A. Camina.  \newblock A survey of the automorphism groups of block
designs. \newblock {\em J. of Combinatorial designs}, { 2:
70-100}, 1994.

{ 4:17-47}, 1972.

\bibitem{Zagier} D. Zagier, \newblock Values of zeta functions and their applications.\newblock {\em  in: A. Joseph
et al. (Eds.), First European Congress of Mathematics}, Vol. II
(Paris, 1992), Progr. Math. 120, Birkh��auser,  Basel, pp.
497-512, 1994.


\bibitem{Bollobas} B{\'e}la Bollob{\'a}s.  \newblock Almost every
  graph has reconstruction number three.  \newblock {\em J. Graph
    Theory}, 14(1):1--4, 1990.

\bibitem{WikipediaReconstruction} Wikipedia contributors.  \newblock
  Reconstruction conjecture.  \newblock {\em Wikipedia, the free
    encyclopedia}, 2011.



\bibitem{HHRT} Edith Hemaspaandra, Lane~A. Hemaspaandra,
  Stanis{\l}aw~P. Radziszowski, and Rahul Tripathi.  \newblock
  Complexity results in graph reconstruction.  \newblock {\em Discrete
    Appl. Math.}, 155(2):103--118, 2007.

\bibitem{Kelly} Paul~J. Kelly.  \newblock A congruence theorem for
  trees.  \newblock {\em Pacific J. Math.}, 7:961--968, 1957.

\bibitem{KSU} Masashi Kiyomi, Toshiki Saitoh, and Ryuhei Uehara.
  \newblock Reconstruction of interval graphs.  \newblock In {\em
    Computing and combinatorics}, volume 5609 of {\em Lecture Notes in
    Comput. Sci.}, pages 106--115. Springer, 2009.

\bibitem{RM} S.~Ramachandran and S.~Monikandan.  \newblock Graph
  reconstruction conjecture: reductions using complement, connectivity
  and distance.  \newblock {\em Bull. Inst. Combin. Appl.},
  56:103--108, 2009.

\bibitem{RR} David Rivshin and Stanis{\l}aw~P. Radziszowski.
  \newblock The vertex and edge graph reconstruction numbers of small
  graphs.  \newblock {\em Australas. J. Combin.}, 45:175--188, 2009.

\bibitem{Stockmeyer} Paul~K. Stockmeyer.  \newblock The falsity of the
  reconstruction conjecture for tournaments.  \newblock {\em J. Graph
    Theory}, 1(1):19--25, 1977.

\bibitem{Ulam} S.~M. Ulam.  \newblock {\em A collection of
    mathematical problems}.  \newblock Interscience Tracts in Pure and
  Applied Mathematics, no. 8.  Interscience Publishers, New
  York-London, 1960.

\end{thebibliography}
